



Songqin Zhao<sup>1</sup>, Diyun Peng<sup>1</sup>, Huwei Wen<sup>1,2,\*</sup> and Huilin Song<sup>3</sup>

- <sup>1</sup> School of Economics and Management, Nanchang University, Nanchang 330031, China
- <sup>2</sup> Research Center of the Central China for Economic and Social Development, Nanchang University, Nanchang 330031, China
- <sup>3</sup> School of International Economics and Trade, Jiangxi University of Finance and Economics, Nanchang 330013, China
- \* Correspondence: wenhuwei@ncu.edu.cn

**Abstract:** As the driving force for the upgrading of the global industrial structure, digital technology has been at the helm of transforming the current economic and technological paradigm. This study empirically analyzes the role of the digital economy in the upgrading industrial structure using panel data from 237 prefecture-level cities in China from 2011–2019. Empirical results show that the development of the digital economy has a significant positive and direct effect on the upgrading of industrial structure, as measured by the two indicators of advancement and servitization. In addition, the digital economy significantly promotes upgrading of the technological progress and human capital of Chinese cities, thus promoting upgrading of the industrial structure. Moreover, the spatial autoregressive model (SAR) and the Spatial Durbin model (SDM) both show that the digital economy has a positive spatial spillover effect on upgrading of the industrial structure.

**Keywords:** digital economy development; industrial structure; mediation effect model; spatial Durbin model (SDM)



Citation: Zhao, S.; Peng, D.; Wen, H.; Song, H. Does the Digital Economy Promote Upgrading the Industrial Structure of Chinese Cities? *Sustainability* **2022**, *14*, 10235. https://doi.org/10.3390/ su141610235

Received: 19 July 2022 Accepted: 16 August 2022 Published: 17 August 2022

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

# 1. Introduction

Digital technologies have been embedded in every economic and social activity today; nearly all industries now use digital technology to promote production and business reform. A subsequent development is the overturning of traditional business models, which in turn reshapes management, production, and supply chains, leading to new consumption systems able to create manufacturing and service themselves. Notably, digital technology changes market demand; hence, residents can enjoy the convenience of digitalization when consuming new products and services. Improvements in supply efficiency and the resulting changes of market demand lead to changes in the industrial structure. The digital economy is therefore believed to lead to the eventual upgrading of the industrial structure, accelerating the process of industrialization and improving the level of economic service.

China is the world's largest industrial economy. However, following the international financial crisis in 2008 it has been faced with the demand for industrial structure upgrading. On one hand, the rise of factor costs has reduced the cost advantage of product production, which requires supply side industrial structure upgrading. On the other hand, increases in resident incomes have increased the market demand for services, which requires industrial structure upgrading. In addition, the realization of carbon neutrality requires the industrial structure to shift to high-value-added economic activities in order to ease and extend the value chain of products [1,2]. The digital economy provides an impetus to upgrade the industrial structure. As a country with this advantage, China urgently needs to use digital technology to upgrade said industrial structure. According to the White Paper on Global Digital Economy Development Report released by the China Academy of Information and Communications Technology, the digital economy grew from USD 0.32 trillion to

USD 6.03 trillion dollars from 2005 to 2020, accounting for 14.2–38.6%, respectively, of the country's Gross Domestic Product (GDP), with its growth rate reaching 9.6% in 2020. Furthermore, the penetration rates of the digital economy of the primary, secondary, and tertiary industries are 8.9%, 21%, and 40.7%, respectively. Therefore, China's promotion of industrial structure upgrading through the digital economy is deeply pivotal for other emerging and developing countries.

The digital economy is a new and crucial force in revolutionizing scientific and technological assets, modernizing the economic system, and providing sustainable economic and environmental development, all of which China needs to promote industrial structure reform. This can be achieved in three ways. First, digital technologies and the digital economy are primary drivers of the scientific and technological revolution, which is crucial in enhancing market competition in these areas [3,4]. Second, the digital economy promotes the construction of a modern economic system. With high innovation, strong penetration, and wide coverage, the digital economy takes data as its key production factor, and is therefore both a new economic growth point and a fulcrum that optimizes and upgrades traditional industries, thereby becoming an important engine in constructing a new economic system [5,6]. Finally, in addition to the supply side, the digital economy changes the structure of demand. For example, residents are more willing to enjoy intelligent services of industrial products, with virtual reality technology providing various new entertainment items.

Several studies have discussed the role of information technology or the digital economy in changing basic industrial process and organizational activities, improving the efficiency of resource allocation, and providing industrial structure optimization [7,8]. Other studies have examined how the digital economy affects the structural changes in primary, secondary, and tertiary industry. The existing literature mainly examines the impact of the digital economy on productivity in manufacturing, services, or other specific industries, often using a single digital economy index [9,10]. For example, studies have found that the digital economy promotes green and innovative development in the industrial economy [11,12]. Other studies discuss the spatial spillover effects of the digital economy on the industrial structure [6]. The digital economy and industrial structure both have important spatial spillover characteristics, and ignoring these may greatly reduce the policy implications of research conclusions.

Digital technology is a general-purpose technology leading the fourth industrial revolution, and is having a profound impact on industrial development and its structure. However, the relationship between technology and industry is complex. Does the digital economy lead the industrial structure to change towards the goal of social development? For example, it is arguably better able to meet the service needs of residents and increase the industrial competitiveness of developing countries. Therefore, it is necessary to reveal the laws and characteristics of the digital economy and its contribution to upgrading of the industrial structure. This study empirically investigates the impact of the digital economy on the industrial structure in China in order to discover the new path and theoretical basis of the transformation and upgrading of the industrial structure and sustainable economic development. First, in contrast with existing research, which primarily uses internet penetration, digitalization, and digital finance to stand in for the digital economy, this study constructs a comprehensive index of digital economic development from four dimensions, namely, digital industrialization, industrial digitalization, digital finance, and digital infrastructure, drawing on data from 237 prefecture-level cities in China from 2010–2019. Next, this study's research perspective focuses on both the impact of the digital economy on the upgrading of the industrial structure and on the mechanisms of technological progress and human capital upgrading. Finally, this research adopts the spatial econometric model to explore the spillover effect of the digital economy on the advancement and servitization of the industrial structure by decomposing its direct and indirect effects.

The remainder of the paper is organized as follows: Section 2 provides a brief literature review; Section 3 explains the study's theoretical analysis and research hypothesis; Section 4 presents the methodology used, including model setting, data description, and data collection; and Section 5 outlines and discusses the empirical results. The final section, Section 6, concludes and discusses policy implications.

# 2. Review of Related Literature

Previous studies have posited that the digital economy mainly refers to the information economy or network economy, which are respectively driven by information and communication technologies (ICT) [13]. ICT has led to relatively revolutionary developments in both the economic and societal arenas, and the network economy and digital economy both represent new economic formats gradually formed with the widespread adoption of ICT. In 1996, Tapscott first put forward the concept of the digital economy; Kim defined the digital economy as a new aspect of the economy beyond the existing traditional economic imagination, and research into the digital economy has received more and more attention [14,15]. The Organization for Economic Co-operation and Development (OECD) and the US Census Bureau and Telecommunication Technology Committee (TTC) in Japan have defined the concept of digital economy as well. Several scholars have studied the measurement of digital economy indicators in different dimensions. For example, Barefoot et al. [16] constructed a comprehensive measure of the digital economy using its contribution to GDP. Bukht and Heeks [17] developed a definition estimated from three scopes of relevance: the digital sector, its output based on digital goods and services, and the digitalized economy.

With the rapid development of the digital economy, scholars have begun to focus on exploring the economic and social dividends of the digital economy in an attempt to reveal the positive impact of digital economy on economic growth, social progress, sustainable development, and other aspects. Awad and Albaity [18] revealed that, in addition to the direct contribution of ICT to growth, ICT indirectly promotes economic growth through investment, openness, and education. Other studies have disclosed that digital technology provides a means to help the economy achieve sustainable development [19,20] as well as that digital facilities increase energy demand and pose additional challenges to sustainability [21]. The impact of ICT adoption on the economic performance of enterprises is the most widely discussed topic. Most studies have affirmed the economic benefits of ICT investment, while others studies suggest that a productivity paradox exists [22,23]. Although certain results remain controversial, most researchers believe that the digital economy promotes industrial development and economic efficiency, which is in turn reflected in one or more aspects of financial, environmental, and social performance [24].

On the adjustment of industrial structure, two main dimensions arise, namely, the internal (i.e., industrial structural optimization) and the external characteristics (i.e., industrial structural upgrading). The main method reflecting industrial structure optimization considers the deviation in the output value and employment structure of each industry along with each industry's difference in economic status [25]. As digitalization improves the efficiency of information transmission and promotes the opening of the regional economy [26], it improves the efficiency of resource allocation and promotes the optimization of industrial structure. According to Clark's law, industrial structure upgrading refers to the process of improving the industrial structure from a low level to a high one, including the advancement and servitization of industrial structure.

Studies have measured industrial structure advancement by the hierarchical coefficient of industrial structure [27,28]. Such studies involve the evolution of industrial proportional relations and the improvement of labor productivity [29], or simply use the ratio of added value to GDP in secondary industry and tertiary industry [30]. Digital technology dramatically affects the industrial structure of major industrialized countries, upgrading their industrial structure to the tertiary industry level. Studies have used the ratio of the added

value in the service sector to the added value in the manufacturing sector as an indicator of this industrial structure upgrading [27,31,32].

Industrial structure is closely related to total factor productivity, energy efficiency, economic security, and green sustainable development [33–35], which is why many studies empirically examine the economic, environmental, and social effects of industrial structure. For example, Luan et al. [36] found that industrial structure adjustment can act as a positive measure to reduce energy intensity in China. The modernization of industrial structure is the premise of a country's industrial security and international competitiveness [37], and developing countries have great demand for industrial structure upgrading. Meanwhile, scholars have studied the path of optimizing and upgrading the industrial structure from different dimensions, such as the effectiveness of progress in capital-embodied technology, green innovation, and digital technologies in upgrading industrial structures [38,39]. Moreover, environmental regulation, ecological compensation, and low-carbon policies have been shown to have a positive effect on upgrading of the industrial structure [25,40,41].

Considering the wide application of digital technology in various industries, studies have investigated the relationship between digital technologies and industrial structure. Many studies have investigated the influence of digital technologies on specific industries. Park and Heo [9] comprehensively examined the impacts of ICT convergence on value chain changes in the electricity industry, focusing on changes in the value chain. Del et al. [10] revealed that ICT plays a positive role in improving the performance and overall financial stability of the banking industry. Other studies have explored the digital economy and industrial structure. For example, Kim and Park [42] examined the crucial role of the ICT industry in the global network based on Korean industry, showing the significance of emphasizing the outflow of ICT industrial knowledge to promote information with respect to other non-ICT industries. Li [43] found that technical change and internet development greatly reduced labor costs and led to industrial structure by developing the digital economy, and explored the relationship between scientific and technological innovation and industrial structure upgrading.

#### 3. Theoretical Analysis and Research Hypothesis

#### 3.1. The Direct Impact of the Digital Economy on the Industrial Structure

The development of the digital economy is accompanied by artificial intelligence, blockchain, 5G, cloud computing, big data, and the internet of things (IoT). Its penetration of various industries has broken the limitations of time and space, resulting in reduced communication costs and expanded markets for products, along with other factors. Hence, digitalization can optimize the production process, efficiently connect all links in the industrial chain, create an intelligent economic ecosystem, and help enterprises to realize innovation in new production processes, products, and business models [44]. Manufacturers can improve their financial performance by introducing product add-on services, and consumers can enjoy better service. The digital economy therefore induces changes in the supply and demand structure, leading to upgrading of the industrial structure.

Specifically, this includes three aspects. First, digital industrialization promotes the new business models of digital industries, improves the development of service industries, and cultivates new industries based on information and communication, big data, software technologies, and more [45]. For example, the virtual reality industry can be applied in various service industries to promote productivity improvements and scale expansion. Second, industrial digitalization can be integrated with primary, secondary, and tertiary industries to improve the mechanization and informatization of primary industry, enhance the interconnection and intelligence of secondary industry, and enrich the diversification and customization of tertiary industry, thereby improving the efficiency of production, creating new requirements, and upgrading the industrial structure [34]. This not only promotes industrial economic development, it promotes the industrial extension of manufacturers to the service industry. Third, the digital economy changes the structure of

market demand. In this era of the digital economy, consumers are more willing to accept the services of intelligent products, and the digital economy promotes the demand for entertainment and culture-related services. The structural change in market demand is the main driving force for the business adjustment of profit-seeking enterprises. In the digital economy era, the industrial economy spontaneously adapts to the increased market demand for services [46]. Based on these factors, we propose the following hypothesis.

**Hypothesis 1 (H1).** *The development of the digital economy can significantly promote the advancement and servitization of the industrial structure.* 

## 3.2. The Mediation Effect of the Digital Economy on the Industrial Structure

The digital economy can expand the enterprise boundary by making it easier to exchange knowledge and information inwards and outwards and to acquire, share, and create new knowledge and information. This is advantageous in promoting innovation and improving efficiency. As a result, innovation can promote the transformation and upgrading of all links in the industrial chain and then motivate industry to achieve leapfrog upgrading [47,48]. Technological progress is the continuous driving force of industrial structure upgrading [49,50]. It is important both to improve the economic and environmental performance of industry and to promote the development of economic services through productivity improvements. Moreover, data (as the new factor) can reshape the traditional factors and improve the efficiency of resource allocation, leading the digital economy to increase the capacity of human capital to enjoy the upgrading of industrial structure by improving the matching degree of employment skills and industrial technology [51]. The development of the digital economy therefore improves the return on human capital, thus attracting more highly skilled talent, and the cycle repeats. Based on these factors, we propose the following hypothesis.

**Hypothesis 2 (H2).** *The digital economy promotes industrial structure upgrading through the intermediary mechanism of technological progress and human capital upgrading.* 

## 3.3. The Spatial Spillover of the Digital Economy on the Industrial Structure

According to Zhao et al. [52], the upgrading of the industrial structure has significant spatial spillover effects. The digital economy weakens the space and time constraints of information transmission and expands the effective circulation of data elements, which in turn strengthens the depth and breadth of coordinated economic development between regions. With the digitalization of industry, enterprises in the industrial chain can share knowledge, the convenience of technology and management is improved, and cooperation between the upstream and downstream industrial chains among regions can be strengthened [19]. In addition, digital finance can meet the demands of consumers and enterprises in different regions, promoting the flow and sharing of elements between regions by foregoing geophysical limitations, thus improving the effect of resource allocation and upgrading the industrial structure between regions [53]. Furthermore, the development of digital finance is conducive to increasing e-commerce and enhancing the level of tertiary industry. The digital economy has intensified competition among regions. It has a network effect, which may cause high-end industries to gather in developed areas. Based on this premise, we propose the following hypothesis.

**Hypothesis 3 (H3).** The development of the digital economy has a special spillover effect on industrial structure upgrading.

# 4. Model Setting and Data Description

# 4.1. Model Setting

# 4.1.1. Panel Data Model

To test the relationship between the development of the digital economy and the industrial structure, this paper adopted Ordinary Least Squares (OLS) with clustering robust standard deviation under time and individual dual fixed effects, expressed herein as Equation (1):

$$lnIndustry\_str_{i,t} = \alpha_0 + \alpha_1 lnDigital_{i,t} + \alpha_m lnX_{i,t} + \eta_t + \mu_i + \varepsilon_{i,t}$$
(1)

where *i* and *t* represent prefectural-level cities and time, respectively; *lnIndustry\_str* is the dependent variable, which was measured from two dimensions, the advancement of industrial structure (*lnIndustry\_gh*) and the servitization of industrial structure (*lnIndustry\_ser*); *lnDigital* is the digital economy development index; *X* is a series of control variables;  $\eta_t$  is a time fixed effect;  $u_i$  is an individual fixed effect;  $\varepsilon_{it}$  is a random perturbation term; and  $\alpha_1$  and  $\alpha_m$  are the coefficients to be estimated. In order to control for the heteroscedasticity and multicollinearity of the model, this study used the natural logarithmic form for all the explanatory variables.

### 4.1.2. Mediation Effect Model

The digital economy may affect the industrial structure through technological progress and human capital. To study the potential indirect impacts of the digital economy on the advancement and servitization of the industrial structure, we adopted the mediation effect model to carry out further empirical investigation:

$$lnMediation_{i,t} = b_0 + b_1 lnDigital_{i,t} + \alpha_m lnX_{i,t} + \eta_t + \mu_i + \varepsilon_{i,t}$$
(2)

$$lnIndustry\_str_{i,t} = c_0 + c_1 lnDigital_{i,t} + \lambda lnMediation_{i,t} + \alpha_m lnX_{i,t} + \eta_t + \mu_i + \varepsilon_{i,t}$$
(3)

where *lnMediation* represents the mediation variables, including the technological progress (*lnRD*) and human capital (*lnHR*), and  $b_1$  represents the effect of the digital economy on mediation variables. Among the mediating variables,  $c_1$  represents the direct effect of the digital economy on the industrial structure,  $\lambda$  represents the effect of mediating variables on the industrial structure,  $b_0$ ,  $c_0$  represents the coefficient constant, and the other variables are the same as in Equation (1).

#### 4.1.3. Spatial Econometric Model

We selected Moran's I index to test the spatial autocorrelation before conducting our empirical analysis. The formula for Moran's I is as follows:

$$Moran's I = \frac{n}{\sum_{i=1}^{n} \sum_{j=1}^{n} W_{ij}} \times \frac{\sum_{i=1}^{n} \sum_{j=1}^{n} W_{ij}(y_i - \overline{Y})(y_i - \overline{Y})}{\sum_{i=1}^{n} (y_i - \overline{Y})^2} = \frac{n \sum_{i=1}^{n} \sum_{j=1}^{n} W_{ij}(y_i - \overline{Y})(y_i - \overline{Y})}{S^2 \sum_{i=1}^{n} \sum_{j=1}^{n} W_{ij}}$$
(4)

where  $S^2 = \sum_{i=1}^n (y_i - \overline{Y})^2$  and  $\overline{Y} = \frac{1}{n} \sum_{i=1}^n y_i$ .

We tested the spatial correlation between cities using the local Moran's I, for which the formula is as follows:  $I_i = z_i \times \sum_{j=1}^n w_{ij}z_j$ , where  $z_i = y_i - \overline{Y}$  and  $z_j = y_j - \overline{Y}$  are the deviation between the observed value and the mean, respectively.

The industrial structure and development of the digital economy in China show spatial correlation characteristics [52,54]. This paper adopted the spatial econometric model to study the relationship between the digital economy and industrial structure. However, the approach of the main spatial econometric models to the applicable objects is quite different. The main spatial econometric models include a spatial autoregressive model (SAR), a spatial error model (SEM), and a spatial Durbin model (SDM). Elhorst [55] found that spatial interaction effects must be analyzed before building a spatial econometric model.

A spatial autoregressive model (SAR) is mainly used to analyze whether the explained variable has a spatial spillover effect after involving the spatial lag variable of the explained variable. This model is expressed below:

$$lnIndustry\_str_{i,t} = \alpha_0 + \rho \sum_{j=1}^{N} W_{ij}lnIndustry\_str_{i,t} + \beta_1 lnDigital_{i,t} + \delta_i X_{i,t} + \mu_i + \varepsilon_{i,t}$$
(5)

where *i* and *t* represent prefecture-level cities and time, respectively; *lnIndustry\_str* is the dependent variable; *lnDigital* represents the explanatory variable for digital economy development; *X* is a series of control variables;  $\rho$  is the spatial spillover coefficient of the industrial structure;  $\alpha_0$ ,  $\beta_1$ , and  $\delta_i$  are series of coefficients;  $\mu_i$  is a city fixed effect;  $\varepsilon_{it}$  is a random perturbation term; and  $W_{ij}$  is an  $N \times N$  order spatial weight matrix.

The Spatial Error Model (SEM) mainly analyzes the spatial effects of missing variables or unobservable random shocks (Yang et al., 2018). The SEM model is constructed as follows:

$$lnIndustry_{stri,t} = \alpha_0 + \beta_1 lnDigital_{i,t} + \delta_i X_{i,t} + u_{i,t}$$
(6)

$$u_{i,t} = \lambda \sum_{j=1}^{N} W_{ij} u_{i,t} + \varepsilon_{i,t}$$
(7)

where  $\lambda$  is the spatial error coefficient.

The SDM is a synthesis between the aSAR model and SEM, with more general results in practical applications. The SDM includes both endogenous and exogenous interaction effects, with each controlling the spatial effects of the explanatory variables, which makes the parameter estimation results more robust. We constructed the SDM as follows:

$$lnIndustry\_str_{i,t} = \alpha_0 + \rho_1 \sum_{j=1}^{N} W_{ij}lnIndustry\_str_{i,t} + \beta_1 lnDigital_{i,t} + \rho_2 \sum_{j=1}^{N} W_{ij}lnDigital_{i,t} + \sigma_i X_{i,t} + \rho_i \sum_{i=1}^{N} W_{ij} X_{i,t} + \mu_i + \varepsilon_{i,t}$$

$$(8)$$

where *i* and *t* represent prefecture-level cities and time, respectively; *lnIndustry\_str* is the dependent variable; *lnDigital* represents the explanatory variable for digital economy development; *X* is a series of control variables; and  $\rho$  is a series of the SAR coefficient. Other variables are similar, as explained above.

In this study, a 0–1 rook spatial weight matrix and inverse distance geographic matrix were used to measure the spatial spillover effect. The 0–1 rook spatial weight matrix ( $W_1$ ) is defined as  $W_{i,j} = \begin{cases} 1, i \neq j \\ 0, i = j \end{cases}$ , where prefecture-level city *i* has a common boundary with city *j*. Then,  $W_{ij} = 1$ ; otherwise,  $W_{ij} = 0$ . The inverse distance geographic matrix ( $W_2$ ) is defined as  $W_{ij} = \begin{cases} \frac{1}{d_{ij}}, i \neq j \\ 0, i = j \end{cases}$ , where  $d_{ij}$  is the surface distance of the prefectural-level city as calculated by its latitude and longitude.

#### 4.2. Data Description

#### 4.2.1. Industrial Structure

Following Gong et al. [56], we constructed an industrial structure advancement indicator that reveals both the shift trend from primary industry to secondary and tertiary industry and the change trend from secondary to tertiary industry. The detailed measurement of industrial structure advancement (*Industry\_gh*) was conducted as follows:

$$Industry\_gh_{i,j,t} = \sqrt{(q_{i,2,t} + q_{i,3,t}) \left(q_{i,2,t} + \frac{q_{i,3,t}}{q_{i,1,t} + q_{i,2,t}}\right)}$$
(9)

here, j = 1, 2, 3 and  $q_{i,j,t}$  represents the ratio of the output value of different industry levels to the regional GDP, with  $q_{i,1,t} + q_{i,2,t} + q_{i,3,t} = 1$ .

Referring to the servitization of the industrial structure according to Zheng et al. [27], we measured this indicator through the ratio of tertiary to secondary industry,  $Industry\_ser_{i,t} = q_{i,3,t}/q_{i,2,t}$ .

# 4.2.2. Comprehensive Digital Economy Development

Digital economy development (*lnDigital*) is a new and relatively complex systematic concept. Therefore, a single simple indicator such as internet development cannot sufficiently reflect the development of China's actual digital economy. Thus, we built a comprehensive indicator system to reflect the current level of China's digital economy development in combination with the development of digital finance [28] (see Table 1). The present study uses an objective weighting method to accurately estimate objects based on the information entropy principle, and uses time variables for a reasonable analysis of the digital economy index.

| Target Level                           | Standard Level            | Index Level                          | Index Interpretation                                                          |  |
|----------------------------------------|---------------------------|--------------------------------------|-------------------------------------------------------------------------------|--|
|                                        | Digital infractory styres | Mobile infrastructure                | Total number of mobile phones                                                 |  |
|                                        | Digital infrastructure    | Internet infrastructure              | Total number of internet users                                                |  |
|                                        | Digital industrialization | Employment of digital industries     | Employees in information transmission, software, and information technology   |  |
| Comprehensive digital<br>economy index | 0                         | Business scale of digital industries | Total sales value of telecom service                                          |  |
| ,, ,                                   | Industry digitalization   | The market of digital business       | Total value of e-commerce sales                                               |  |
|                                        |                           | The coverage of digital finance      | The coverage breadth of digital finance                                       |  |
|                                        | Digital finance           | The usage depth of digital finance   | The index of payment, credit, insurance, credit, investment, and money funds. |  |
|                                        |                           | The digitization in finance          | The digitization level of digital finance                                     |  |

Table 1. Evaluation system of digital economy comprehensive index.

#### 4.2.3. Mediating Variables

We selected both technological progress (*lnRD*) and human capital (*lnHR*) as moderating variables. Using investment in research and development as a proxy for technological progress, technological progress was found to improve resource utilization and production efficiency, thus affecting the upgrading of the industrial structure [6]. Modelled as the number of university students per 10,000 people in the region [57], human capital (*lnHR*) significantly improved the use of resource elements, enhanced capacity for technological absorption, and even promoted upgrading of the industrial structure [5].

#### 4.2.4. Control Variables and Data Sources

To minimize errors in the regression results from the omission of variables, the following control variables were selected for the models. Economic development (*lnGDP*) was expressed as the growth speed of regional GDP. Foreign direct investment (*lnFDI*) stimulates the upgrading of industrial structure in local industries by bringing in new capital and advanced technologies. The cities' public finance budget expenditure (*lnGovern-ment*), total fixed assets invested (*lnInvestment*), deposits from residents (*lnFinance*), and the proportion of the urban population with respect to the local permanent population (*lnURB*) all promote changes to the industrial structure [58].

All data for these indicators were from the following sources: the China Environmental Statistical Yearbook, Statistical Report on China's Internet Development Status, China Financial Statistics Yearbook, China Statistical Yearbook on Science and Technology, and the National Bureau of Statistics. Relevant missing data were filled in by interpolation. Table 2 presents the descriptive statistics for the variables used in this study. The LLC test is the panel unit root test, and certain variables report the results of the trend unit root test as well. This shows that the results of the panel unit-root test (LLC Test) reject the null hypothesis at the 1% significance level, indicating that these variables are stationary or trend stationary.

| Variables      | S.D.  | Mean   | Min    | Max    | Correlation1 | Correlation2 | LLC Test    |
|----------------|-------|--------|--------|--------|--------------|--------------|-------------|
| lnIndustry_gh  | 0.144 | 0.025  | -0.550 | 0.826  | 1.000        | -            | -2.087 ***  |
| lnIndustry_ser | 0.464 | -0.141 | -1.743 | 1.643  | -            | 1.000        | -6.492 ***  |
| lnDigital      | 0.559 | -2.607 | -4.437 | -0.388 | 0.632 ***    | 0.566 ***    | -12.974 *** |
| lnGDP          | 0.933 | 7.426  | 5.231  | 10.549 | 0.570 ***    | 0.198 ***    | -4.098 ***  |
| lnFDI          | 2.012 | 10.058 | 0.000  | 14.705 | 0.418 ***    | 0.083 ***    | -18.698 *** |
| lnGovernment   | 0.772 | 14.900 | 12.031 | 18.241 | 0.530 ***    | 0.390 ***    | -9.306 ***  |
| lnFinance      | 2.522 | 8.277  | 5.309  | 20.495 | 0.419 ***    | 0.381 ***    | -5.445 ***  |
| lnInvestment   | 0.914 | 7.151  | 2.732  | 9.792  | 0.509 ***    | 0.173 ***    | -36.912 *** |
| lnURB          | 0.280 | 3.977  | 0.311  | 5.929  | 0.523 ***    | 0.206 ***    | -13.959 *** |
| lnRD           | 1.412 | 1.242  | -2.622 | 6.775  | 0.581 ***    | 0.228 ***    | -7.193 ***  |
| lnHR           | 1.000 | 4.756  | 0.737  | 7.147  | 0.462 ***    | 0.163 ***    | -37.382 *** |

Table 2. Descriptive statistics.

Note: \*\*\* indicates significance at 1% level. Correlation refers to the correlation coefficients between industrial structure and other variables.

#### 5. Empirical Results and Discussion

#### 5.1. Estimated Results of Baseline Regression

To make the results more comparable, columns (1) and (3) list the results of the panel model under city fixed effect, while columns (2) and (4) show the results of the panel model under time and city fixed effects. According to Table 3, the results of the Hausman test are significant at the 5% and 1% level, while the likelihood ratio (LR) test results are all significant at the 1% level, which means that the results can be regressed with clustering robust standard deviation under time and city fixed effects.

**Table 3.** The direct impact of digital economy development on industrial structure.

| Variables               | (1)           | (2)                    | (3)            | (4)                      |
|-------------------------|---------------|------------------------|----------------|--------------------------|
|                         | lnIndustry_gh | lnIndustry_gh          | lnIndustry_ser | lnIndustry_ser           |
| lnDigital               | 0.015 **      | 0.019 **               | 0.211 ***      | 0.127 ***                |
|                         | (0.007)       | (0.008)                | (0.037)        | (0.027)                  |
| lnGDP                   | 0.226 ***     | 0.143 ***              | 0.803 ***      | 0.313                    |
|                         | (0.036)       | (0.041)                | (0.281)        | (0.333)                  |
| lnFDI                   | 0.001         | 0.003                  | 0.000          | 0.012 *                  |
|                         | (0.002)       | (0.002)                | (0.008)        | (0.006)                  |
| lnGovernment            | 0.035 ***     | 0.012                  | 0.226 ***      | 0.054                    |
|                         | (0.009)       | (0.009)                | (0.064)        | (0.068)                  |
| lnFinance               | 0.001 *       | -0.001                 | -0.000         | -0.009 ***               |
|                         | (0.000)       | (0.001)                | (0.002)        | (0.003)                  |
| lnInvestment            | 0.003         | 0.013 *                | 0.014          | 0.033                    |
|                         | (0.007)       | (0.007)                | (0.022)        | (0.022)                  |
| lnURB                   | 0.043         | 0.028                  | 0.185          | 0.086                    |
|                         | (0.028)       | (0.021)                | (0.143)        | (0.084)                  |
| Hausman test<br>LR test |               | 15.82 **<br>336.95 *** |                | 147.81 ***<br>510.91 *** |
| City fixed effect       | Yes           | Yes                    | Yes            | Yes                      |
| Time fixed effect       | No            | Yes                    | No             | Yes                      |
| R-squared               | 0.735         | 0.774                  | 0.823          | 0.861                    |
| Observations            | 2133          | 2133                   | 2133           | 2133                     |

Note: The prefix "*ln*" before the explanatory variables denotes a logarithmic form. \*\*\*, \*\*, and \* indicate significance at the 1%, 5%, and 10% levels, respectively. Figures in () are standard errors.

Empirical results show that the direct effect of digital economy development on the advancement of industrial structure is significantly positive at the 5% level, while the direct effect on the servitization of the industrial structure is significant at the 1% level. This indicates that the development of the digital economy can promote the three levels of industry (primary, secondary, and tertiary), enhancing the advancement and serviti-

zation of the industrial structure and thereby proving the first hypothesis. Two reasons can explain these findings: first, the integration of digital technologies occurs mainly in secondary and tertiary industry; for example, in 2020, digital industrialization and industrial digitization accounted for 7.3% and 31.2% of GDP, respectively, with the ratio even reaching 40.7% for the service industry. Second, the existence of many new models and forms of business emerging in the consumption and service sectors breaks space and time limitations, promoting the development of a new service industry and the servitization of digital industry.

Moreover, the regression results of other factors basically meet the theoretical expectations. With economic development (*lnGDP*) and investment (*lnInvestment*) having positive effects on the upgrading of the industrial structure, the newly increased economic activity in Chinese cities mainly comes from advanced or service-oriented industries. Foreign direct investment (*lnFDI*) can stimulate enhancement of servitization in the industrial structure. However, higher deposits from residents (*lnFinance*) leads to reduced spending, which hinders the development of tertiary industry. The impact of urbanization on industrial structure remains uncertain.

# 5.2. Estimated Results of the Mediation Effect Model

Table 4 reports the estimation results of the mediation effect model using technological progress (lnRD) and human capital upgrading (lnHR) as mediation variables. Columns (1) to (3) use technological progress as the mediation variable, while columns (4) to (6) use human capital upgrading. Columns (1) and (4) show that the effect coefficients of the digital economy on technological progress (0.109) and human capital upgrading (0.165) are significantly positive, indicating that the digital economy can promote technological progress and enhance human capital levels.

| Variables                                         | (1)           | (2)                    | (3)                    | (4)           | (5)                    | (6)                    |
|---------------------------------------------------|---------------|------------------------|------------------------|---------------|------------------------|------------------------|
| variables =                                       | lnRD          | lnIndustry_gh          | lnIndustry_ser         | lnHR          | lnIndustry_gh          | lnIndustry_ser         |
| lnDigital                                         | 0.109 **      | 0.087 ***              | 0.152 ***              | 0.165 ***     | 0.086 ***              | 0.151 ***              |
|                                                   | (0.046)       | (0.007)                | (0.012)                | (0.059)       | (0.007)                | (0.012)                |
| lnGDP                                             | 0.017         | 0.142 ***              | 1.402 ***              | 0.850 ***     | 0.121 ***              | 1.384 ***              |
|                                                   | (0.098)       | (0.015)                | (0.025)                | (0.126)       | (0.015)                | (0.025)                |
| lnFDI                                             | 0.156 ***     | -0.001                 | -0.000                 | 0.063 ***     | 0.001                  | 0.002                  |
|                                                   | -0.011        | (0.002)                | (0.003)                | (0.014)       | (0.002)                | (0.003)                |
| lnGovernment                                      | 0.682 ***     | -0.071 ***             | 0.020                  | -0.577 ***    | -0.038 ***             | 0.047 ***              |
|                                                   | (0.056)       | (0.009)                | (0.015)                | (0.072)       | (0.008)                | (0.015)                |
| lnFinance                                         | 0.003         | 0.002 **               | 0.000                  | 0.007         | 0.002 **               | 0.001                  |
|                                                   | (0.007)       | (0.001)                | (0.002)                | (0.009)       | (0.001)                | (0.002)                |
| lnInvestment                                      | 0.070 *       | 0.006                  | 0.022 **               | 0.214 ***     | 0.003                  | 0.019 *                |
|                                                   | (0.039)       | (0.006)                | (0.010)                | (0.051)       | (0.006)                | (0.010)                |
| lnURB                                             | 0.588 ***     | 0.095 ***              | 0.142 ***              | 0.453 ***     | 0.099 ***              | 0.146 ***              |
|                                                   | (0.068)       | (0.010)                | (0.018)                | (0.087)       | (0.010)                | (0.017)                |
| lnRD                                              |               | 0.027 ***<br>(0.003)   | 0.022 ***<br>(0.006)   |               |                        |                        |
| lnHR                                              |               |                        |                        |               | 0.025 ***<br>(0.002)   | 0.021 ***<br>(0.004)   |
| Indirect effect                                   |               | 0.003 **<br>(0.001)    | 0.002 **<br>(0.001)    |               | 0.004 ***<br>(0.002)   | 0.004 **<br>(0.001)    |
| Sobel_z value<br>Observations<br>Adjust R-squared | 2133<br>0.779 | 2.273<br>2133<br>0.527 | 2.021<br>2133<br>0.866 | 2133<br>0.268 | 2.676<br>2133<br>0.535 | 2.422<br>2133<br>0.866 |

Table 4. Estimated results of mediation effect.

Note: The prefix "*ln*" before the explanatory variables denotes a logarithmic form. \*\*\*, \*\*, and \* indicate significance at the 1%, 5%, and 10% level, respectively. Figures in () are robust standard errors.

It is evident from columns (2) and (5) that the regression coefficients for the relationship between technological progress, human capital level, and advancement of the industrial structure are all significantly positive at the 1% level, at 0.087 and 0.086, respectively. These results pass the Sobel test, indicating that the digital economy can positively upgrade the industrial structure by promoting technological progress and human capital levels. In light of these results, we can see that the indirect effects of the digital economy on industrial structure advancement through technological progress and human capital level are significantly positive, at 0.003 and 0.004, respectively.

Columns (3) and (6) show that the impact coefficients of technological progress and human capital level between the servitization of industrial structure are all significantly positive at the 1% level, at 0.152 and 0.151, respectively. These results pass the Sobel test as well, with T-values of 2.021 and 2.422, respectively. This indicates that the digital economy can positively promote the development of the industrial structure servitization of the industrial structure is indirectly enhanced by the digital economy, with values of 0.002 and 0.004, respectively.

The two possible reasons for the above-mentioned results are as follows. First, the digital economy can allow entrepreneurs to more conveniently obtain innovative resources, greatly improving innovation efficiency [6]. Digital technology can be applied to new industries, such as new chips, basic software, intelligent hardware, etc., to promote industrial modernization [39]. The digital economy can integrate with traditional industries, promote the intelligent development of traditional industries, and push enterprises to manufacture products for consumers and provide full life-cycle services. Moreover, as digitalization needs highly educated and skilled labor, human capital improves through increased sharing of knowledge among platforms and easier access to knowledge, improving management and technological progress and promoting the advancement of the industrial structure. Nevertheless, the digital economy may lead to wage polarization and income inequality, which decreases its potential to increase industrial efficiency and impedes upgrading of the industrial structure [59].

# 5.3. Estimated Results of the Spatial Durbin Model

# 5.3.1. Spatial Correlation Test

In this study, Moran's I index is used to evaluate the spatial correlation of industrial structure in various Chinese regions. Table 5 shows that the global Moran's I index values for the advancement of industrial structure (*lnIndustry\_gh*), the servitization of industrial structure (*lnIndustry\_ser*), and digital economy development (*lnDigital*) from 2011–2019 are significantly positive at the 1% level. The null hypothesis of no spatial autocorrelation is significant; therefore, China's industrial structure and digital economy development have significant spatial autocorrelation, and conducting spatial econometric analysis is appropriate.

|      |       | lnDigital |                 |       | lnIndustry_g | h               | l     | nIndustry_s | er              |
|------|-------|-----------|-----------------|-------|--------------|-----------------|-------|-------------|-----------------|
| Time | Ι     | Z         | <i>p</i> -Value | Ι     | Z            | <i>p</i> -Value | Ι     | Z           | <i>p</i> -Value |
| 2011 | 0.253 | 5.943     | 0.000           | 0.323 | 7.556        | 0.000           | 0.133 | 3.164       | 0.001           |
| 2012 | 0.216 | 5.095     | 0.000           | 0.335 | 7.823        | 0.000           | 0.140 | 3.308       | 0.000           |
| 2013 | 0.184 | 4.381     | 0.000           | 0.329 | 7.678        | 0.000           | 0.154 | 3.638       | 0.000           |
| 2014 | 0.191 | 4.557     | 0.000           | 0.323 | 7.55         | 0.000           | 0.205 | 4815        | 0.000           |
| 2015 | 0.162 | 3.886     | 0.000           | 0.327 | 7.647        | 0.000           | 0.243 | 5.686       | 0.000           |
| 2016 | 0.145 | 3.505     | 0.000           | 0.309 | 7.233        | 0.000           | 0.231 | 5.406       | 0.000           |
| 2017 | 0.136 | 3.274     | 0.001           | 0.304 | 7.116        | 0.000           | 0.209 | 4.906       | 0.000           |
| 2018 | 0.117 | 2.819     | 0.002           | 0.271 | 6.361        | 0.000           | 0.197 | 4.615       | 0.000           |
| 2019 | 0.190 | 4.470     | 0.000           | 0.221 | 5.211        | 0.000           | 0.180 | 4.239       | 0.000           |

Table 5. Moran's index for industrial structure and digital economy development.

### 5.3.2. Model Selection Test

Table 6 shows the diagnostic test results for the spatial econometric model under the 0–1 rook spatial weight matrix ( $W_1$ ) and the inverse distance geographic matrix ( $W_2$ ). For advancement of the industrial structure, the value of the Lagrange multiplier (LM) test, robust Lagrange multiplier (LM) test, and Wald test are all positive. Therefore, the choice of the spatial Durbin model is reasonable. Meanwhile, for the servitization of the industrial structure, we found that the values for the robust LM (lag) test are insignificant. Hence, we report the results for the SAR model instead.

| Fable 6. Diagnostic | test results | for spatial | model |
|---------------------|--------------|-------------|-------|
|---------------------|--------------|-------------|-------|

| Test                    | <i>lnIndu</i><br>(und | <i>stry_gh</i><br>er W <sub>1</sub> ) | <i>lnIndu</i><br>(und | estry_gh<br>er W2) | <i>lnIndu</i><br>(unde | stry_ser<br>er W1) | <i>lnIndu</i><br>(unde | stry_ser<br>er W2) |
|-------------------------|-----------------------|---------------------------------------|-----------------------|--------------------|------------------------|--------------------|------------------------|--------------------|
|                         | Value                 | <i>p</i> -Value                       | Value                 | <i>p</i> -Value    | Value                  | <i>p</i> -Value    | Value                  | <i>p</i> -Value    |
| LM (error) test         | 33.548                | 0.000                                 | 28.535                | 0.000              | 11.975                 | 0.001              | 24.425                 | 0.000              |
| Robust LM (error) test  | 21.049                | 0.000                                 | 3.255                 | 0.071              | 10.470                 | 0.001              | 17.780                 | 0.000              |
| LM (lag) test           | 15.333                | 0.000                                 | 28.656                | 0.000              | 1.545                  | 0.214              | 8.207                  | 0.004              |
| Robust LM (lag) test    | 2.833                 | 0.092                                 | 3.272                 | 0.066              | 0.040                  | 0.842              | 0.562                  | 0.454              |
| Wald test spatial lag   | 47.290                | 0.000                                 | 82.800                | 0.000              | 168.960                | 0.000              | 176.870                | 0.000              |
| Wald test spatial error | 90.560                | 0.000                                 | 38.390                | 0.000              | 104.930                | 0.000              | 166.230                | 0.000              |

# 5.3.3. Spatial Effect Estimation Results

Table 7 reports the estimation results of the spatial econometric models. Columns (1) and (2) report the estimation results of the digital economy on advancement of the industrial structure based on SAR under  $W_1$  and  $W_2$ . The regression coefficients all are positive, with 0.011 at the 5% level. Columns (3) and (4) show the results of servitization of the industrial structure based on SDM under  $W_1$  and  $W_2$ . Evidently, the coefficients are all positive, at 0.140 and 0.105 at the 1% level. These results are consistent with the estimated coefficients for the basic models in Table 3, showing the robustness of our spatial econometric models. This indicates that the development of the digital economy can significantly upgrade the industrial structure.

Table 7. Estimation results of spatial econometric models.

|              | lnIndu             | stry_gh            | lnIndus            | stry_ser           |
|--------------|--------------------|--------------------|--------------------|--------------------|
| Variables –  | (1) W <sub>1</sub> | (2) W <sub>2</sub> | (3) W <sub>1</sub> | (4) W <sub>2</sub> |
| lnDigital    | 0.011 **           | 0.011 **           | 0.140 ***          | 0.105 ***          |
|              | (0.005)            | (0.005)            | (0.013)            | (0.013)            |
| lnGDP        | 0.167 ***          | 0.135 ***          | 0.581 ***          | 0.311 ***          |
|              | (0.012)            | (0.012)            | (0.038)            | (0.039)            |
| lnFDI        | 0.001              | 0.002 ***          | 0.001              | 0.010 ***          |
|              | (0.001)            | (0.001)            | (0.003)            | (0.003)            |
| lnGovernment | 0.026 ***          | 0.017 ***          | 0.155 ***          | 0.061 ***          |
|              | (0.006)            | (0.006)            | (0.020)            | (0.020)            |
| InFinance    | 0.002 ***          | 0.000              | -0.002 **          | -0.009 ***         |
|              | (0.001)            | (0.001)            | (0.001)            | (0.001)            |
| lnInvestment | 0.000              | 0.007 **           | 0.010              | 0.024 **           |
|              | (0.003)            | (0.003)            | (0.010)            | (0.010)            |
| lnURB        | 0.035 ***          | 0.029 ***          | 0.138 ***          | 0.089 ***          |
|              | (0.005)            | (0.005)            | (0.019)            | (0.018)            |
| lnRD         | 0.006 ***          | 0.006 ***          | 0.012 **           | 0.018 ***          |
|              | (0.002)            | (0.002)            | (0.006)            | (0.006)            |
| lnHR         | 0.001              | 0.003 *            | -0.010 **          | 0.003              |
|              | (0.001)            | (0.001)            | (0.005)            | (0.004)            |

| Variables –      | lnIndu             | stry_gh            | lnIndustry_ser     |                    |  |
|------------------|--------------------|--------------------|--------------------|--------------------|--|
|                  | (1) W <sub>1</sub> | (2) W <sub>2</sub> | (3) W <sub>1</sub> | (4) W <sub>2</sub> |  |
| rho              | 0.378 ***          | 0.831 ***          | 0.354 ***          | 0.783 ***          |  |
|                  | (0.024)            | (0.051)            | (0.018)            | (0.030)            |  |
| sigma2_e         | 0.001 ***          | 0.010 ***          | 0.010 ***          | 0.010 ***          |  |
|                  | (0.000)            | (0.000)            | (0.000)            | (0.000)            |  |
| Observations     | 2133               | 2133               | 2133               | 2133               |  |
| R-squared        | 0.150              | 0.419              | 0.515              | 0.229              |  |
| Number of cities | 237                | 237                | 237                | 237                |  |

Table 7. Cont.

Note: The prefix "*ln*" before the explanatory variables denotes a logarithmic form. \*\*\*, \*\*, and \* indicate significance at the 1%, 5%, and 10% level, respectively. Figures in () are standard errors.

To further analyze the effect of digital economy development on regional industrial structure, this paper deconstructs the impact of the digital economy on industrial structure into both direct and spillover effects under  $W_1$  and  $W_2$  based on SDM for advancement of the industrial structure and SAR for servitization of the industrial structure, shown in Table 8. Looking at the results of columns (1) and (2), the coefficients of the direct effect of the digital economy on the advancement of the industrial structure are significantly positive at 0.010. However, the coefficients of the industrial structure of local regions, it impedes upgrading of the industrial structure in neighboring regions. In columns (3) and (4), the coefficients of the direct and indirect effect based on SAR under  $W_1$  and  $W_2$  are all significantly positive at the 1% level. This indicates that the development of the digital economy can upgrade the industrial structure of the digital economy can enhance the level of industrial structure servitization in both local and neighboring regions.

| lnIndustry_ser     |                      |  |
|--------------------|----------------------|--|
| (3) W <sub>1</sub> | (4) W <sub>2</sub>   |  |
| 0.144 ***          | 0.107 ***            |  |
| (0.014)            | (0.013)              |  |
| 0.071 ***          | 0.385 ***            |  |
| (0.007)            | (0.069)              |  |
| 0.215 ***          | 0.491 ***            |  |
| (0.019)            | (0.074)              |  |
|                    | 0.215 ***<br>(0.019) |  |

Table 8. Estimation results of decomposition effects.

Note: The prefix "*ln*" before the explanatory variables denotes a logarithmic form. \*\*\*, \*\*, and \* indicate significance at the 1%, 5%, and 10% level, respectively. Figures in () are standard errors.

In essence, the development of the digital economy promotes the advancement and servitization of the industrial structure in local regions and enhances the servitization of the industrial structure of neighboring ones. However, it hinders upgrading of the industrial structure in neighboring regions. This can be explained by the following reasons. First, the development of the digital economy can promote the development of technological progress and human capital, which upgrades the industrial structure and enhances the servitization of the industrial structure in local regions. Next, the digital economy can strengthen the advantages of the local industrial structure, which promotes further upgrading within the local industrial structure from labor-intensive and capital-intensive industries to technology-intensive and green-intensive ones. Hence, it impedes upgrading of the industrial structure of neighboring regions. Third, the development of digital servitization gives rise to new business models, which both enriches the variety of tertiary

industries and expands the scope of service, bringing an increasing level of tertiary industry to both local and neighboring areas.

#### 5.4. Estimated Results of Robustness Analysis

The 2SLS method is a feasible method for solving problems related to endogenous explanatory variables [60]. As the lagged terms of the independent variables satisfy the exogeneity and correlation conditions, they are usually selected as instrumental variables. In this study, we chose the first-order lag term of the development of digital economy as the instrumental variable. In order to avoid measurement deviation in the digital economy indicators, we adopted broadband infrastructure as a proxy variable for the digital economy. According to Zhou et al. [61], broadband is a modern infrastructure serving the digital economy. In 2014, China officially selected pilot cities to implement China's broadband strategy and increase their construction of broadband infrastructure. This paper defines these pilot cities as the treatment group, other non-pilot cities as the control group, and uses the difference-in-differences (DID) design to evaluate the causal relationship between the digital economy and industrial structure.

Finally, from the perspective of the three major industries, the upgrading of industrial structure means that the proportion of primary industry continues to decline, the proportion of secondary industry first rises then subsequently falls, and the proportion of tertiary industry continues to rise. To avoid the influence of deviation in the index measures on our estimation results, we refer to Zheng [27] in replacing the calculation method for the industrial structure in our robustness test. This study calculates the industrial structure, expressed as  $Industry\_ser_1 = q_1 \times 1 + q_2 \times 2 + q_3 \times 3$ , where  $q_i$  represents the ratio of output value of different industry to the regional GDP, i = 1, 2, 3.

Table 9 presents the results of the linear regressions. Columns (1) and (3) report the results of the advancement and servitization of industrial structure under 2SLS, showing that the coefficients are all significantly positive, at 0.240 and 0.265 at the 1% level. Columns (2) and (4) report the results of DID estimation, showing that the Broadband City strategy significantly upgrades the industrial structure and promotes servitization at the 1% level, reconfirming the reliability of our findings. Columns (5) and (6) report the results of OLS without control variables, while columns (7) and (8) show the results of OLS with control variables. As the results of the Hausman and LR tests are all significant at the 1% level, this study adopts the Ordinary Least Squares (OLS) with clustering robust standard deviation under time and individual dual fixed effects in columns (6) and (8). It is clear that the coefficients of digital economy on upgrading of the industrial structure are significantly positive, at 0.018 and 0.021 at the 1% level. This indicates the robustness of positive effect of digital economy on upgrading structure.

Moreover, Table 10 reports the results of the robustness test of the mediation effect with technological progress and human capital, showing that the coefficients of the mediation effects are at 0.001 and 0.001 at the 1% level of significance. The robustness results of the spatial effect are shown in Table 11, showing that the direct effects are significant at 0.012 and 0.013 under  $W_1$  and  $W_2$ . However, the indirect effects are significantly negative at -0.008 and -0.193, respectively. Therefore, these results indicate that while the development of the digital economy can upgrade the industrial structure in local regions by prompting technological progress and human capital, it impedes the same development in neighboring regions. This matches the results in Tables 4 and 8, demonstrating the robustness of the mediation effect and spatial effect between the digital economy and upgrading of the industrial structure.

|                                                                     | lnIndus                    | stry_gh                     | lnIndus                    | stry_ser                    |                            | lnIndus                     | try_ser1                   |                             |
|---------------------------------------------------------------------|----------------------------|-----------------------------|----------------------------|-----------------------------|----------------------------|-----------------------------|----------------------------|-----------------------------|
| Variables                                                           | (1)                        | (2)                         | (3)                        | (4)                         | (5)                        | (6)                         | (7)                        | (8)                         |
| lnDigital                                                           | 0.016 ***<br>(0.005)       | 0.240 ***<br>(0.020)        | 0.124 ***<br>(0.028)       | 0.265 ***<br>(0.053)        | 0.052 ***<br>(0.001)       | 0.018 ***<br>(0.004)        | 0.020 ***<br>(0.003)       | 0.021 ***<br>(0.003)        |
| lnGDP                                                               | 0.055 **<br>(0.026)        | 0.083 ***<br>(0.030)        | -0.397 **<br>(0.159)       | 1.312 ***<br>(0.138)        |                            |                             | 0.090 ***<br>(0.018)       | 0.043 **<br>(0.019)         |
| lnFDI                                                               | -0.003<br>(0.003)          | 0.001<br>(0.002)            | -0.031 ***<br>(0.012)      | 0.003<br>(0.006)            |                            |                             | 0.001<br>(0.001)           | 0.002 *<br>(0.001)          |
| lnGov                                                               | 0.087 ***<br>(0.023)       | -0.072 ***<br>(0.013)       | 0.697 ***<br>(0.148)       | 0.018<br>(0.046)            |                            |                             | 0.015 ***<br>(0.005)       | 0.000<br>(0.004)            |
| lnFin                                                               | 0.004 ***<br>(0.001)       | -0.001<br>(0.002)           | 0.021 ***<br>(0.003)       | 0.001<br>(0.002)            |                            |                             | -0.001<br>(0.001)          | -0.001 ***<br>(0.001)       |
| lnInvest                                                            | 0.003<br>(0.008)           | 0.034 ***<br>(0.008)        | 0.015<br>(0.049)           | 0.034 **<br>(0.017)         |                            |                             | 0.003<br>(0.003)           | 0.007 ***<br>(0.003)        |
| InURB                                                               | 0.059 *<br>(0.032)         | 0.080 ***<br>(0.020)        | 0.340 *<br>(0.186)         | 0.127 ***<br>(0.029)        |                            |                             | 0.022 *<br>(0.013)         | 0.013<br>(0.008)            |
| Constant                                                            | -1.938 ***<br>(0.161)      | 0.843 ***<br>(0.222)        | -8.919 ***<br>(0.992)      | -1.267 ***<br>(0.347)       | 0.963 ***<br>(0.003)       | 0.858 ***<br>(0.014)        | 0.375 ***<br>(0.084)       | 0.759 ***<br>(0.090)        |
| City fixed effect<br>Time fixed effect<br>Observations<br>R-squared | Yes<br>No<br>2133<br>0.650 | Yes<br>Yes<br>1896<br>0.529 | Yes<br>No<br>2133<br>0.588 | Yes<br>Yes<br>1896<br>0.862 | Yes<br>No<br>2133<br>0.674 | Yes<br>Yes<br>2133<br>0.779 | Yes<br>No<br>2133<br>0.765 | Yes<br>Yes<br>2133<br>0.810 |

## Table 9. Robustness test of linear regression.

Note: The prefix "*ln*" before the explanatory variables denotes a logarithmic form. \*\*\*, \*\*, and \* indicate significance at the 1%, 5%, and 10% level, respectively. Figures in () are standard errors.

#### Table 10. Robustness test of mediation effect.

| Variables                                         | (1)                    | (2)                    | (4)                  | (5)                    |
|---------------------------------------------------|------------------------|------------------------|----------------------|------------------------|
|                                                   | lnRD                   | $lnIndustry\_ser_1$    | lnHR                 | $lnIndustry\_ser_1$    |
| lnDigital                                         | 0.109 **<br>(0.046)    | 0.041 ***<br>(0.003)   | 0.165 ***<br>(0.059) | 0.041 ***<br>(0.003)   |
| lnRD                                              |                        | 0.011 ***<br>(0.001)   |                      |                        |
| lnHR                                              |                        |                        |                      | 0.009 ***<br>(0.001)   |
| Constant                                          | -15.398 ***<br>(0.777) | 0.939 ***<br>(0.052)   | 5.766 ***<br>(1.000) | 0.719 ***<br>(0.048)   |
| Indirect effect                                   |                        | 0.001 **<br>(0.001)    |                      | 0.001 ***<br>(0.001)   |
| Sobel_z value<br>Observations<br>Adjust R-squared | 2133<br>0.779          | 2.275<br>2133<br>0.578 | 2133<br>0.268        | 2.634<br>2133<br>0.578 |

Note: The prefix "*ln*" before the explanatory variables denotes a logarithmic form. \*\*\* and \*\* indicate significance at the 1% and 5% level, respectively. Figures in () are robust standard errors.

## Table 11. Robustness test of spatial effect.

| Variable                  | Spatial Weight Matrix: W <sub>1</sub> |                      |                        |                     | Spatial Weight Matrix: W <sub>2</sub> |                      |                        |                     |
|---------------------------|---------------------------------------|----------------------|------------------------|---------------------|---------------------------------------|----------------------|------------------------|---------------------|
|                           | (1) Main<br>Effect                    | (2) Direct<br>Effect | (3) Indirect<br>Effect | (4) Total<br>Effect | (5) Main<br>Effect                    | (6) Direct<br>Effect | (7) Indirect<br>Effect | (8) Total<br>Effect |
| lnDigital                 | 0.012 ***<br>(0.002)                  | 0.012 ***<br>(0.002) | -0.008 *<br>(0.004)    | 0.004<br>(0.004)    | 0.014 ***<br>(0.002)                  | 0.013 ***<br>(0.002) | -0.193 *<br>(0.108)    | -0.179 *<br>(0.109) |
| rho                       | 0.449 ***<br>(0.022)                  |                      |                        |                     | 0.879 ***<br>(0.038)                  |                      |                        |                     |
| sigma2_e                  | 0.000 ***<br>(0.000)                  |                      |                        |                     | 0.000 ***<br>(0.000)                  |                      |                        |                     |
| Observations<br>R-squared | 2133<br>0.023                         | 2133<br>0.023        | 2133<br>0.023          | 2133<br>0.023       | 2133<br>0.269                         | 2133<br>0.269        | 2133<br>0.269          | 2133<br>0.269       |
| Number of<br>cities       | 237                                   | 237                  | 237                    | 237                 | 237                                   | 237                  | 237                    | 237                 |

Note: The prefix "ln" before the explanatory variables denotes a logarithmic form. \*\*\* and \* indicate significance at the 1% and 10% level, respectively. Figures in () are standard errors.

# 6. Conclusions and Policy Recommendations

The challenge of climate change has brought external requirements to the upgrading of the industrial structure [62], while digitalization has brought internal impetus to the same. This study constructs a comprehensive measurement system for the digital economy and then investigates the impact of the digital economy on the industrial structure based on panel data from 237 prefectural-level cities in China from 2011–2019. It adopts panel regression to empirically analyze the direct impact of the digital economy on upgrading the industrial structure, and uses a mediation model to prove the mediation effect of technological progress and human capital upgrading. Furthermore, this study reveals the spatial spillover effect of the digital economy on the advancement and servitization of the industrial structure based on the SDM and SAR models.

Our main research conclusions are as follows. First, development of the digital economy can promote the transformation and adjustment of industrial structure in local regions from a low-level state to a high-level one. Robustness tests, such as the method of calculating the digital economy and industrial structure, that is, the 2SLS method, show that this finding is robust. Second, the digital economy can promote upgrading of the urban industrial structure through the intermediary mechanisms of promoting technological progress and human capital upgrading. Finally, there exists a spatial spillover effect between the digital economy and industrial structure upgrading. Specifically, development of the digital economy can promote the servitization of the industrial structure in a region and its surrounding areas. Although the digital economy can promote the advancement of the industrial structure in this region, however, it hinders advancement of the industrial structure in surrounding areas.

Based on these findings, the following policy recommendations are proposed. All regions can take advantage of the digital technology dividend to promote industrial upgrading and service orientation. With the declining international demand for manufacturing and the country's rising internal production costs, China's industrial structure faces an urgent need for transformation and upgrading. In addition to improving digital infrastructure, developing countries need to formulate industrial policies for the digital economy in order to help industries realize advanced and service-oriented upgrading through digital transformation. The effectiveness of the industrial policy for development of the digital economy deserves attention, as it is the key to determining a region's competitive advantage in the opportunity period of global industrial structure adjustment [63,64]. Moreover, it is important for enterprises to improve the knowledge, experience, and skills of their human resources and more effectively match their current pool with their organizational needs through development of the digital economy in order to improve production efficiency, management efficiency, and strategic development. Finally, the government should pay attention to the spatial spillover effect of the digital economy on industrial structure upgrading and strengthen spatial cooperation in industrial upgrading [65]. In the era of the digital economy, regional competition and cooperation between industries are more important. The government should formulate regional coordination policies for digital economic development in order to promote cooperation in industrial development.

This paper presents a preliminary discussion about the relationship between the digital economy and the industrial structure; however, there are deficiencies in this study and more remains to be done. First, it would be better to use an integrated index for both the digital economy and industry in order to reveal the effect of development of the digital economy on the industrial structure. However, we were only able to construct a comprehensive index of the digital economy, as we lacked official statistics for an integrated index. Second, while this paper has tried to investigate the impact of the digital economy on the industrial structure, our results do not reflect the effects on the internal structure of manufacturing. The upgrading of the internal structure of manufacturing in the digital economy era is one of the topics that must be studied in the future. In addition, studies in specific regions may be helpful in providing more targeted policy implications.

**Author Contributions:** Conceptualization by D.P. and H.W.; methodology by S.Z. and H.W.; data curation by S.Z. and H.S.; writing—original draft by S.Z.; writing—review and editing by H.W; supervision by D.P. and H.W. All authors have read and agreed to the published version of the manuscript.

**Funding:** This research was funded by the Youth Program of the National Social Science Foundation of China (20CJY031).

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

**Data Availability Statement:** The data presented in this study are available on request from the corresponding author.

Conflicts of Interest: The authors declare no conflict of interest.

#### References

- Kaivo-oja, J.; Luukkanen, J.; Panula-Ontto, J.; Vehmas, J.; Chen, Y.; Mikkonen, S.; Auffermann, B. Are structural change and modernisation leading to convergence in the CO<sub>2</sub> economy? Decomposition analysis of China, EU and USA. *Energy* 2014, 72, 115–125. [CrossRef]
- 2. Kahrl, F.; Roland-Holst, D. Growth and structural change in China's energy economy. Energy 2009, 34, 894–903. [CrossRef]
- 3. Ma, D.; Zhu, Q. Innovation in emerging economies: Research on the digital economy driving high-quality green development. *J. Bus. Res.* **2022**, *145*, 801–813. [CrossRef]
- 4. Benitez, J.; Arenas, A.; Castillo, A.; Esteves, J. Impact of digital leadership capability on innovation performance: The role of platform digitization capability. *Inf. Manag.* 2022, *59*, 103590. [CrossRef]
- Li, Y. The Influence of the Development of Digital Economy on the Upgrading of China's Industrial Structure. E3S Web Conf. 2021, 235, 03062. [CrossRef]
- 6. Su, J.; Su, K.; Wang, S. Does the Digital Economy Promote Industrial Structural Upgrading?—A Test of Mediating Effects Based on Heterogeneous Technological Innovation. *Sustainability* **2021**, *13*, 10105. [CrossRef]
- Qian, W.; Liu, H.; Pan, F. Digital Economy, Industry Heterogeneity, and Service Industry Resource Allocation. Sustainability 2022, 14, 8020. [CrossRef]
- D'Ippolito, B.; Petruzzelli, A.M.; Panniello, U. Archetypes of incumbents' strategic responses to digital innovation. J. Intellect. Cap. 2019, 20, 662–679. [CrossRef]
- 9. Park, C.; Heo, W. Review of the changing electricity industry value chain in the ICT convergence era. *J. Clean. Prod.* **2020**, 258, 120743. [CrossRef]
- 10. Del Gaudio, B.L.; Porzio, C.; Sampagnaro, G.; Verdoliva, V. How do mobile, internet and ICT diffusion affect the banking industry? An empirical analysis. *Eur. Manag. J.* **2021**, *39*, 327–332. [CrossRef]
- 11. Wen, H.; Lee, C.C.; Song, Z. Digitalization and environment: How does ICT affect enterprise environmental performance? *Environ. Sci. Pollut. Res.* **2021**, *28*, 54826–54841. [CrossRef] [PubMed]
- 12. Wen, H.; Zhong, Q.; Lee, C.C. Digitalization, competition strategy and corporate innovation: Evidence from Chinese manufacturing listed companies. *Int. Rev. Financ. Anal.* **2022**, *82*, 102166. [CrossRef]
- 13. Seo, H.J.; Lee, Y.S.; Oh, J.H. Does ICT investment widen the growth gap? Telecommun. Policy 2009, 33, 422-431. [CrossRef]
- 14. Tapscott, D. Six Themes for New Learning from: The Digital Economy: Promise and Peril in the Age of Networked Intelligence. *Educom Rev.* **1996**, *31*, 52–54.
- 15. Kim, J. Infrastructure of the digital economy: Some empirical findings with the case of Korea. *Technol. Forecast. Soc. Chang.* **2006**, 73, 377–389. [CrossRef]
- 16. Barefoot, K.; Curtis, D.; Jolliff, W.; Nicholson, J.R.; Omohundro, R. *Defining and Measuring the Digital Economy*; US Department of Commerce Bureau of Economic Analysis: Washington, DC, USA, 2018; p. 15.
- 17. Bukht, R.; Heeks, R. *Defining, Conceptualising and Measuring the Digital Economy*. Development Informatics Working Paper No. 68. 2017. Available online: https://papers.ssrn.com/sol3/papers.cfm?abstract\_id=3431732 (accessed on 15 March 2022).
- Awad, A.; Albaity, M. ICT and economic growth in Sub-Saharan Africa: Transmission channels and effects. *Telecommun. Policy* 2022, 102381. [CrossRef]
- 19. Zafar, M.W.; Zaidi, S.A.H.; Mansoor, S.; Sinha, A.; Qin, Q. ICT and education as determinants of environmental quality: The role of financial development in selected Asian countries. *Technol. Forecast. Soc. Chang.* **2022**, *177*, 121547. [CrossRef]
- 20. Chatti, W.; Majeed, M.T. Information communication technology (ICT), smart urbanization, and environmental quality: Evidence from a panel of developing and developed economies. *J. Clean. Prod.* **2022**, *366*, 132925. [CrossRef]
- 21. Lange, S.; Pohl, J.; Santarius, T. Digitalization and energy consumption. Does ICT reduce energy demand? *Ecol. Econ.* **2020**, 176, 106760. [CrossRef]
- 22. Cuevas-Vargas, H.; Parga-Montoya, N.; Fernández-Escobedo, R. The adoption of ICT as an enabler of frugal innovation to achieve customer satisfaction. The mediating effect of frugal innovation. *Procedia Comput. Sci.* 2022, 199, 198–206. [CrossRef]

- Karim, M.S.; Nahar, S.; Demirbag, M. Resource-Based Perspective on ICT Use and Firm Performance: A Meta-analysis Investigating the Moderating Role of Cross-Country ICT Development Status. *Technol. Forecast. Soc. Chang.* 2022, 179, 121626. [CrossRef]
- 24. Hao, Y.; Guo, Y.; Wu, H. The role of information and communication technology on green total factor energy efficiency: Does environmental regulation work? *Bus. Strategy Environ.* **2022**, *31*, 403–424. [CrossRef]
- Zheng, Q.; Wan, L.; Wang, S.; Wang, C.; Fang, W. Does ecological compensation have a spillover effect on industrial structure upgrading? Evidence from China based on a multi-stage dynamic DID approach. *J. Environ. Manag.* 2021, 294, 112934. [CrossRef] [PubMed]
- 26. Zhou, F.; Wen, H.; Lee, C.C. Broadband infrastructure and export growth. Telecommun. Policy 2022, 46, 102347. [CrossRef]
- 27. Zheng, J.; Shao, X.; Liu, W.; Kong, J.; Zuo, G. The impact of the pilot program on industrial structure upgrading in low-carbon cities. *J. Clean. Prod.* 2021, 290, 125868. [CrossRef]
- Guo, F.; Wang, J.; Wang, F.; Kong, T.; Zhang, X.; Cheng, Z. Measuring China's digital financial inclusion: Index compilation and spatial characteristics. *China Econ. Q.* 2020, 19, 1401–1418.
- Wang, L.; Wang, Z.; Ma, Y. Heterogeneous environmental regulation and industrial structure upgrading: Evidence from China. Environ. Sci. Pollut. Res. 2022, 29, 13369–13385. [CrossRef]
- Ma, B.; Yu, Y. Industrial structure, energy-saving regulations and energy intensity: Evidence from Chinese cities. *J. Clean. Prod.* 2017, 141, 1539–1547. [CrossRef]
- 31. Chong, Z.; Qin, C.; Ye, X. Environmental regulation and industrial structure change in China: Integrating spatial and social network analysis. *Sustainability* **2017**, *9*, 1465. [CrossRef]
- Lu, X.H.; Jiang, X.; Gong, M.Q. How land transfer marketization influence on green total factor productivity from the approach of industrial structure? Evidence from China. *Land Use Policy* 2020, 95, 104610. [CrossRef]
- Li, K.; Lin, B. Economic growth model, structural transformation, and green productivity in China. *Appl. Energy* 2017, 187, 489–500. [CrossRef]
- 34. Li, D.; Chen, Y.; Miao, J. Does ICT create a new driving force for manufacturing?—Evidence from Chinese manufacturing firms. *Telecommun. Policy* **2022**, *46*, 102229. [CrossRef]
- 35. Su, Y.; Fan, Q.M. Renewable energy technology innovation, industrial structure upgrading and green development from the perspective of China's provinces. *Technol. Forecast. Soc. Chang.* **2022**, *180*, 121727. [CrossRef]
- 36. Luan, B.; Zou, H.; Chen, S.; Huang, J. The effect of industrial structure adjustment on China's energy intensity: Evidence from linear and nonlinear analysis. *Energy* **2021**, *218*, 119517. [CrossRef]
- 37. Mingaleva, Z.; Gataullina, A. Structural modernization of economy and aspects of economic security of territory. *Middle East J. Sci. Res.* **2012**, *12*, 1535–1540. [CrossRef]
- 38. You, J.; Zhang, W. How heterogeneous technological progress promotes industrial structure upgrading and industrial carbon efficiency? Evidence from China's industries. *Energy* **2022**, 247, 123386. [CrossRef]
- 39. Mingaleva, Z. Structural modernization of economy and innovation development. World Appl. Sci. J. 2012, 20, 1313–1316.
- 40. Chen, X.; Qian, W. Effect of marine environmental regulation on the industrial structure adjustment of manufacturing industry: An empirical analysis of China's eleven coastal provinces. *Mar. Policy* **2020**, *113*, 103797. [CrossRef]
- 41. Chen, M. A study of low-carbon development, urban innovation and industrial structure upgrading in China. *Int. J. Low-Carbon Technol.* **2022**, *17*, 185–195. [CrossRef]
- 42. Kim, M.S.; Park, Y. The changing pattern of industrial technology linkage structure of Korea: Did the ICT industry play a role in the 1980s and 1990s? *Technol. Forecast. Soc. Chang.* **2009**, *76*, 688–699. [CrossRef]
- 43. Li, Y. Internet development and structural transformation: Evidence from China. J. Appl. Financ. Bank. 2020, 10, 153–172.
- 44. Miao, Z. Digital economy value chain: Concept, model structure, and mechanism. Appl. Econ. 2021, 53, 4342–4357. [CrossRef]
- 45. Raddats, C.; Naik, P.; Bigdeli, A.Z. Creating value in servitization through digital service innovations. *Ind. Mark. Manag.* 2022, 104, 1–13. [CrossRef]
- Chen, Y.; Wang, L. Commentary: Marketing and the sharing economy: Digital economy and emerging market challenges. *J. Mark.* 2019, *83*, 28–31. [CrossRef]
- 47. Anderson, P.; Tushman, M.L. Technological discontinuities and dominant designs: A cyclical model of technological change. In *Organizational Innovation*, 1st ed.; Routledge: London, UK, 2018; pp. 373–402.
- Urbinati, A.; Chiaroni, D.; Chiesa, V.; Frattini, F. The role of digital technologies in open innovation processes: An exploratory multiple case study analysis. *RD Manag.* 2020, 50, 136–160. [CrossRef]
- 49. Ghasemaghaei, M.; Calic, G. Assessing the impact of big data on firm innovation performance: Big data is not always better data. *J. Bus. Res.* **2020**, *108*, 147–162. [CrossRef]
- 50. Du, K.; Cheng, Y.; Yao, X. Environmental regulation, green technology innovation, and industrial structure upgrading: The road to the green transformation of Chinese cities. *Energy Econ.* **2021**, *98*, 105247. [CrossRef]
- 51. Michaels, G.; Natraj, A.; Van Reenen, J. Has ICT polarized skill demand? Evidence from eleven countries over twenty-five years. *Rev. Econ. Stat.* **2014**, *96*, 60–77. [CrossRef]
- 52. Zhao, J.; Jiang, Q.; Dong, X.; Dong, K.; Jiang, H. How does industrial structure adjustment reduce CO<sub>2</sub> emissions? Spatial and mediation effects analysis for China. *Energy Econ.* **2022**, *105*, 105704. [CrossRef]

- 53. Wu, Y.; Huang, S. The effects of digital finance and financial constraint on financial performance: Firm-level evidence from China's new energy enterprises. *Energy Econ.* **2022**, *112*, 106158. [CrossRef]
- 54. Li, X.; Liu, J.; Ni, P. The Impact of the digital economy on CO<sub>2</sub> emissions: A theoretical and empirical analysis. *Sustainability* **2021**, 13, 7267. [CrossRef]
- 55. Elhorst, J.P. Spatial Econometrics: From Cross-Sectional Data to Spatial Panels; Springer: Berlin/Heidelberg, Germany, 2014; Volume 479. [CrossRef]
- 56. Gong, R.Z.; Pan, F.P.; Liu, Y.X. Research on measurement model of industrial structure supererogation and its properties. *Stat. Decis.* **2021**, *12*, 67–71. (In Chinese)
- Yao, X.; Shah, W.U.H.; Yasmeen, R.; Zhang, Y.; Kamal, M.A.; Khan, A. The impact of trade on energy efficiency in the global value chain: A simultaneous equation approach. *Sci. Total Environ.* 2021, 765, 142759. [CrossRef] [PubMed]
- 58. Jiang, H.; Jiang, P.; Wang, D.; Wu, J. Can smart city construction facilitate green total factor productivity? A quasi-natural experiment based on China's pilot smart city. *Sustain. Cities Soc.* **2021**, *69*, 102809. [CrossRef]
- Ren, S.; Hao, Y.; Xu, L.; Wu, H.; Ba, N. Digitalization and energy: How does internet development affect China's energy consumption? *Energy Econ.* 2021, 98, 105220. [CrossRef]
- 60. Martens, E.P.; Pestman, W.R.; de Boer, A.; Belitser, S.V.; Klungel, O.H. Instrumental variables: Application and limitations. *Epidemiology* **2006**, *17*, 260–267. [CrossRef]
- Tang, C.; Xu, Y.; Hao, Y.; Wu, H.; Xue, Y. What is the role of telecommunications infrastructure construction in green technology innovation? A firm-level analysis for China. *Energy Econ.* 2021, 103, 105576. [CrossRef]
- Wen, H.; Lee, C.C.; Zhou, F. How does fiscal policy uncertainty affect corporate innovation investment? Evidence from China's new energy industry. *Energy Econ.* 2022, 105, 105767. [CrossRef]
- 63. Appiah-Otoo, I.; Song, N. The impact of ICT on economic growth-Comparing rich and poor countries. *Telecommun. Policy* **2021**, 45, 102082. [CrossRef]
- 64. Wen, H.; Zhao, Z. How does China's industrial policy affect firms' R&D investment? Evidence from 'Made in China 2025'. *Appl. Econ.* **2021**, *53*, 6333–6347. [CrossRef]
- 65. Gruber, H. Innovation, skills and investment: A digital industrial policy for Europe. *Econ. Politica Ind.* **2017**, *44*, 327–343. [CrossRef]